Author's response to reviews

Title: Analysis on the equity of government investment to primary healthcare institutions during the 3-year health reform in China

Authors:

Xiaopeng X Zhang Z (yuhuaduola@163.com)
Yuqi Y Xiong X (xiongxiaogi0525@163.com)
Jing J Ye Y (392236734@qq.com)
Zhaohua Z Deng D (642337036@qq.com)
Xinping X Zhang Z (xpzhang602@163.com)

Version: 2 Date: 12 April 2012

Author's response to reviews: see over
Dear Editors:

Thank you very much for your advice. I realize that I haven’t described clearly in our manuscript. In fact, we use the data of monitoring and evaluation of national essential drug system conducted by China’s Ministry of Health in 2011, but I wrote inappropriately in our manuscript before. So I revise the method in our manuscript and upload the revised version. Thanks a lot for informing.

Best Regards.
Yours Sincerely,

Xinping Zhang, Ph.D M.D. Professor in Health Policy and Management
School of Health and Medical Management, Tongji Medical College of Huazhong University of Science and Technology
No.13. Hangkong Road, Wuhan, Hubei province, 430030, China.
Tel: +86 2783692332; Fax: +86 2783692996
E-mail: xpzhang602@163.com